Author's response to reviews

Title: Supported housing programs form person with serious mental illnesses in rural northern communities: A mixed method evaluation

Authors:

Phyllis Montgomery (pmontgomery@laurentian.ca)
Cheryl Forchuk (cforchuk@uwo.ca)
Craig M Duncan (cduncan@laurentian.ca)
Don Rose (donrose@ryerson.ca)
Patricia H Bailey (pbailey@laurentian.ca)
Ramamohan Veluri (rveluri@normed.ca)

Version: 2 Date: 17 July 2008

Author's response to reviews:

Good afternoon,

I received an email from Dr. Liz Hoffman requesting an RCT registration number. The study protocol is not a RCT. Instead, I most likely pressed the wrong keys during the uploading of my manuscript. The method used in this proposal is mixed methods.

My apologies.

Phyllis Montgomery